The effects of school closures on female entrepreneurship during the COVID-19 crisis: Evidence from Chile

January 2nd, 2024

Abstract

This resarch analyses the effects of school closures on female entrepreneurships during the COVID-19 pandemic in Chile. Building on institutional theories of gender inequalities in entrepreneurship, we argue that schools are an important formal institution affecting the entrepreneurial process. Using a rich database of administrative records, we show that school closures significantly decreased the probability of parents of starting a business. We estimate that, due to the uneven distribution of household responsibilities and cultural expectations regarding childcare, the gender gap in entrepreneurship was widened by 29,6%. This gendered impact persisted even after schools reopened. We observed differential effects based on the age of children. Mothers with children aged between 15 and 18 years experienced less pronounced setbacks during school closures and displayed a quicker recovery in returning to pre-COVID levels of entrepreneurial activity following school reopenings compared to mothers with children aged between 3 and 6 years. We also document that women from poorer backgrounds (quintiles I and II) were the most affected from closures compared with women from richer households. Our findings maintain robustness across various model specifications, reinforcing the validity of a causal interpretation. We contribute to the literature on female entrepreneurship with insights from an emerging economy in times of crisis, showing how school closures are an institutional barrier that amplifies entrenched gender-based disparities in entrepreneurship.

Keywords: Entrepreneurship, COVID-19, Schools, female labour participation

JEL Codes: key1, key2, key3

Contents

1.	Int	oduction	. 2
2.	Lite	erature Review	.4
	a.	Formal and informal institutional barriers to entrepreneurship	.4
	b.	Gender and institutional barriers to entrepreneurship	.6
3.	Со	ntext	.9
4.	Мє	thodology1	1
a.	Ι	Data sources1	1
	i.	Registry of Companies of Chile (RES)1	1
	ii.	Social Household Registry (RSH)1	3
	iii.	Civil registry (SRCeI)1	4
	iv.	COVID-19 data1	4
	v.	Educational records1	15
b.	Ι	Descriptive statistics1	18
c.	E	mpirical Strategy	20
	i.	Main regression2	20
	ii.	Heterogenous effects2	21
5.	Re	sults2	22
a.	N	ain Results2	22
b.	ł	leterogeneous impacts2	25
	i.	By children's age2	25
	ii.	By socioeconomic quintiles2	27
6.	Ro	bustness checks2	29
7.	Dis	cussion	31
8.	Со	nclusions	32
9.	Re	ferences	34
10.	A	ppendices4	13
a.	Fig	ures4	ł3
b.	Ta	bles5	53

1. Introduction

The outbreak of COVID-19 created unprecedented challenges across the globe, posing an existential threat to business operations and entrepreneurial activities (Angelov and Waldenström, 2023; Fairlie et al., 2023; Belitski et al., 2022; Brown and Rocha, 2020; Fairlie, 2020; Ratten, 2020). Yet, these challenges have been far from uniform, revealing intricate differences in the experiences of individual businesses and entrepreneurs (Pertuze et al., 2023). This research seeks to uncover one specific and overlooked aspect of this heterogeneity: the differential impact of extended school closures on women's entrepreneurial activities.

Focusing on Chile—an emerging economy that endured one of the lengthiest durations of school closures globally—this study delves into the effects of these interruptions on the registration of businesses owned by both men and women during the pandemic. Recognizing that mothers, often the primary caregivers, might be uniquely impacted by school closures, we employ rich administrative microdata to explore the individual factors that could exacerbate or mitigate these effects. Overall, we predict that school closures may have had a pronounced effect on female entrepreneurship, especially among mothers.

We frame our study through the lens of institutional theory. This theoretical perspective allows us to view school closures as an unexpected institutional barrier that exacerbated already entrenched gender disparities in entrepreneurship. Institutional theory, with its emphasis on understanding how formal rules and informal constraints influence organisational behaviour and outcomes (North, 1990), provides a rich context for analysing the unique challenges faced by female entrepreneurs during the COVID-19 pandemic.

Globally, school closures emerged as a widely adopted non-pharmaceutical intervention to cope with the public health crisis (Hale et al., 2021). This measure affected an estimated 1.6 billion students, originating the largest disruption to education in modern history (Pearson,

2022). Yet the severity and length of these closures differed markedly across and within countries (UNESCO, 2022). For instance, while many nations in Europe came to favour a faster reopening of schools, Latin America and the Caribbean was the region with the highest average number of school days lost (UNESCO, 2022). Chile, in particular, withstood the longest stretch of school closures among the OECD countries, with over 250 days of full closure and 116 days of partial closure (OECD, 2022). That is, most schools were closed for the entire year of 2020 and remained closed for nearly half of 2021, persisting even after vaccines were widely available (Aguilera et al., 2021).

Scholars have emphasised the critical need to evaluate the consequences of school closures during the COVID-19 crisis (Gouëdard et al., 2020; Dasgupta, 2022; Clarke et al.; 2022), particularly on students' development and education (Betthäuser et al., 2023; Agostinelli et al., 2022; Moscoviz and Evans, 2022; Halloran et al., 2021; Kuhfeld et al., 2020). However, while such adverse effects have been extensively documented, the repercussions extend far beyond the classroom. School closures place significant burdens on families, especially in households with working parents. These burdens disproportionately fall on women, who are traditionally tasked with domestic roles and unpaid labour, encompassing activities related to childcare and household management (Fortin, 2005; Gurley-Calvez et al., 2009; Craig et al., 2012; Bertrand, 2018). In Chile, for example, women on average spend 2.9 more hours per day on domestic work than men (World Bank, 2023a). Globally, the COVID-19 crisis exacerbated this unequal distribution of duties (Farré et al., 2020; Del Boca et al., 2020; Sevilla and Smith, 2020; Hipp and Bünning, 2021), with a cross-country study suggesting that women devoted three times as many hours as men to unpaid childcare during the pandemic (Kenny and Yang, 2021).

The objectives and contributions of this paper are manifold. First, developing a novel analytical framework presenting school closures as a distinct institutional barrier within the entrepreneurial process. Second, providing robust empirical evidence on the effects of COVID-19 school closures on entrepreneurial activity in Chile, demonstrating the amplification of entrenched gender inequalities. Third, enriching the literature on female entrepreneurship

with insights from an emerging economy in times of crisis. Finally, influencing future policy responses by underscoring the importance of educational institutions in safeguarding women's entrepreneurial activities and the need for policies that acknowledge gender disparities in entrepreneurship.

The rest of this paper is organised as follows. Section 2 offers a comprehensive literature review, situating our study within existing research on institutional barriers to entrepreneurship and gender disparities. Section 3 explains the different procedures available to start a business in Chile and the responses set by the central government in order to control the pandemic. Section 4 presents the research methodology, outlining the data sources, variables, and econometric techniques used to estimate the effects of school closures on female entrepreneurship. Section 5 details the main empirical results, illustrating how extended school closures have hindered women's entrepreneurial activities. Section 6 discusses the implications of these findings, emphasising the amplification of entrenched gender inequalities and the need for targeted policy responses. Finally, Section 7 concludes by highlighting the study's contributions to the literature on gender and entrepreneurship, suggesting directions for future research, and reflecting on the broader societal implications of school closures to women's business activities.

2. Literature Review

a. Formal and informal institutional barriers to entrepreneurship

Institutional theory (North, 1990) offers an analytical perspective for understanding the complex landscape of entrepreneurship. Through this lens, institutions are seen as stable, predictable patterns guiding human behaviour. Institutions can be divided into two main categories: formal rules, such as laws and regulations, and informal constraints, including cultural norms and social practices (North, 1990).

Institutions act as powerful forces in the market, shaping entrepreneurial activities, strategies, and outcomes (Baumol, 1990; Hwang and Powell, 2005; Bruton et al., 2010; Urbano et al., 2019). Thus, entrepreneurship is an endogenous phenomenon, dynamically influenced by the prevailing institutions within a specific context (Busenitz et al., 2000; Davidsson and Henrekson, 2002; McMullen et al., 2008; Estrin et al., 2013; Urbano and Alvarez, 2014; Boudreaux et al., 2019). The influence of institutions can be multifaceted, either facilitating or hindering entrepreneurial activities, especially in emerging economies or developing countries characterised by capital constraints and high levels of uncertainty (Desai et al., 2003; Welter and Smallbone, 2011). The term 'institutional barrier' denotes any institutional factor that systematically limits certain groups' access to resources, opportunities, or decision-making power.

Institutional barriers may manifest at various levels (Welter, 2011; Thornton et al., 2012). They can shape entrepreneurship on a national scale or affect more localised aspects, such as specific industries, regions, or demographic groups, including women. Understanding how these barriers impact the entrepreneurial process requires a nuanced exploration of both formal and informal barriers, and how these barriers interact with each other.

On the one hand, formal barriers encompass the legal and regulatory frameworks that determine business dynamics (De Soto, 2000; Djankov et al., 2002; Djankov, 2009; Klapper et al., 2006; van Stel et al., 2007; Audretsch and Moog, 2022), from the rules governing market competition (Tirole, 1988; Laffont and Tirole, 2001) to economic and financial policies that can either stimulate or stifle entrepreneurial endeavours (Gompers and Lerner, 1999; Black and Strahan, 2002; Audretsch, 2007; Acs and Szerb, 2007; Minniti, 2008). For instance, burdensome labour regulations can lead to the expansion of the informal economy, where businesses operate outside of legal channels (Botero et al., 2004; Cordier and Bade, 2023).

On the other hand, informal barriers include cultural practices and social norms affecting attitudes towards entrepreneurship (Hayton et al., 2002; Stephan and Uhlaner, 2010; Hayton and Cacciotti, 2013; Autio et al., 2013; Boudreaux et al., 2019). Although less tangible than legal or regulatory hurdles, these factors can influence the entrepreneurial environment by

shaping public perceptions and expectations. From gender roles that may discourage women entrepreneurs to social stigmas against failure that may limit risk-taking, informal barriers operate through subtle cultural mechanisms.

Formal and informal institutional barriers often intertwine. A clear instance of this interplay emerged during the COVID-19 pandemic with school closures. At its core, the directive to close schools stemmed from a formal, legal mandate aimed at curbing the spread of the virus. Yet, it was the deep-seated informal barriers, specifically gender-oriented societal practices and cultural norms, that appear to have disproportionately affected women. Faced with these converging challenges, female entrepreneurs felt compelled to adjust or even sideline their economic activities to assume primary responsibility for their children's immediate needs. Such gender-specific ramifications are not isolated phenomena but rather emanate from a longstanding, systematic arrangement of intersecting institutional barriers that affect women.

b. Gender and institutional barriers to entrepreneurship

Existing literature provides ample evidence of the gendered dimensions of entrepreneurship, often emphasising the role of institutional factors in determining the engagement and performance of women in entrepreneurial ventures (Ahl and Marlow, 2012; Jennings and Brush, 2013; Brush et al., 2019; Foss et al., 2019; Neumeyer et al., 2019; McAdam et al., 2019). Despite international efforts spanning decades, a global disparity persists in business ownership and firm performance, especially in emerging and developing economies (e.g. Hardy and Kagy, 2018; Essers et al., 2020).

Around the globe, only one in three businesses have a woman among the principal owners (World Bank, 2023b). Moreover, women typically earn less in self-employment than both their male counterparts and salaried employees of either gender (Bertrand, 2018). Indeed, women often find themselves pushed into self-employment or small business ownership out

of necessity rather than pulled by the promise of opportunity. Their entrepreneurial ventures frequently gravitate toward customer-oriented services and less productive industries—a trend that might be linked to the more modest growth trajectories and reduced likelihood of survival observed in women-led businesses (Boden Jr. and Nucci, 2000; Coleman, 2007; Coleman and Robb, 2009; Fairlie and Robb, 2009; Kiefer et al., 2022). This lack of representation and performance disparity in entrepreneurship is the outcome of a complex interplay between formal institutions, such as educational systems and labour policies, and informal institutions, including cultural norms and societal expectations. Together, these give rise to differences that further exacerbate the gender gap in entrepreneurial activities, setting the stage for the unique challenges that women encounter in the business sector.

Empirical research has exposed several areas where women face obstacles, such as human, financial, and social capital. In terms of human capital, the evidence indicates that on average women demonstrate lower levels of financial literacy and business acumen than men (Lusardi and Mitchell, 2011; OECD, 2017). This gap is exacerbated by women's self-perceptions, as they often view themselves as less capable of becoming entrepreneurs and hold themselves to higher standards of competence than their male counterparts (Thébaud, 2010). Additionally, societal gender biases consistently undervalue women's leadership qualities (Yang and del Carmen Triana, 2017). Financial capital presents additional challenges, as women tend to experience consistent difficulties in securing funding, an empirical observation that aligns with a broader gender disparity within entrepreneurial finance (Beck et al., 2008; Coleman and Robb, 2009; Agier and Szafarz, 2012; Marlow and McAdam, 2012; Brana, 2013; Eddleston et al., 2016; Lauto et al., 2022). With respect to social capital, not only are women entrepreneurs frequently constrained by limited access to business networks (Allen, 2000; OECD/EU, 2017), but the very quality and character of their connections often exhibit notable differences compared of those of men (Renzulli et al., 2000; Schutjens and Stam, 2003). For instance, finding mentors or sponsors who can guide and support an entrepreneur's career is often more challenging for women.

Cultural norms and societal expectations also play a role in limiting women's entrepreneurial opportunities. Women's motivations within entrepreneurship have been found to be distinct from those of men, often emphasising flexibility and work-life balance (Lombard, 2001; Edwards and Field-Hendrey, 2002; Walker et al., 2008; Craig et al., 2012; Loscocco and Bird, 2012; Thébaud, 2016; Mas and Pallais, 2017). However, the overlap of gender roles with entrepreneurial tasks often complicates achieving a harmonious equilibrium (Hyytinen and Ruskanen, 2007; Brush et al., 2009; Jennings and Brush, 2013). It is noteworthy, however, that assumptions of homogeneity can often obscure and misrepresent the unique challenges that some groups of women encounter (Patrick et al., 2016; Brush and Greene, 2021).

The business environment became increasingly challenging for women in the wake of the COVID-19 pandemic, disproportionately affecting their jobs and entrepreneurial activities (Alon et al., 2020, 2021; Albanesi and Kim, 2021; Graeber et al., 2021; Torres et al., 2023). The imposition of lockdowns and school closures amplified difficulties for employees and entrepreneurs with caregiving responsibilities, a burden predominantly borne by women who typically assume the primary role for childcare (Farré et al., 2020; Del Boca et al., 2020; Sevilla and Smith, 2020; Hipp and Bünning, 2021). While the pandemic intensified these challenges, they are not new. Using data spanning 1994-2019, a recent study by Price & Wasserman (2023) found that summer school closures in the U.S. lead to a 1.1 percentage point decline in women's employment-to-population ratio, a 9.8 percent decrease in total hours worked, and a 3.3 percent drop in women's weekly earnings, emphasising the impact of childcare responsibilities.

Such burdens obviously extend beyond the U.S., taking on additional complexity when considering the intersection of socio-economic status. In Chile, school closures have amplified educational inequalities, with students from less privileged backgrounds affected more severely (Kuzmanic et al., 2023). This pattern suggests that their mothers, already disadvantaged in the labour market, likely faced even greater challenges due to school

closures, possibly experiencing further disruptions in employment or increased childcare responsibilities.

Recent empirical evidence highlights this trend. Data from 50 countries shows that female-led firms were 4 percentage points more likely to close than their male-led counterparts during the pandemic (Goldstein et al., 2022). Similarly, businesses led by women were more prone to suffer extended closures (Liu et al., 2021), and women-owned businesses performed worse due to a higher sensitivity to health and economic risks (Birhanu et al., 2022). This situation is mirrored in academia, where female researchers, especially those with young children, were disproportionately affected (Myers et al., 2021; Deryugina et al., 2021).

3. Context

a. Business registrations in Chile

This study focuses on the first step any formal business must undertake to start operating: the registration of the firm. In Chile, there are two options for legally registering a firm.

First, entrepreneurs can do it through the Official Journal. This option requires entrepreneurs to draft and notarize their company's articles of association, then register the business and publish its details. This entire process, essential for legal business operations, must be completed within 60 days and necessitates the entrepreneur's physical presence. Second, entrepreneurs can opt for the Registry of Companies (*Registro de Empresas y Sociedades*, RES), an online platform introduced on May 2, 2013. The RES was designed to streamline the process of establishing and managing companies, enabling entrepreneurs to draft their articles of association directly online. Before the pandemic, finalising the registration process required paying for a notary's service or buying an Advanced Electronic Signature (AES) from a certified physical store. However, on April 23, 2020, a new online AES became available for nearly 1 US dollar, making the process more accessible and removing the need for in-person interactions.

In this study, we focus on the administrative records from the RES. There are three main reasons. First, being an online platform, its digital nature allows for a more efficient data collection and analysis process, ensuring accuracy and timeliness. Second, since its introduction in 2013, the RES platform progressively became the predominant choice for new entrepreneurs. Third, resulting in a surge of new registrations and a shift to the online method, largely to the detriment of the traditional registration processes (Montégu et al., 2021).

b. School closures in Chile

We aim to understand how school closures and reopenings influenced business registrations from January 2019 to December 2021, and assess how these shifts differently impacted men and women based on their roles as parents.

Chile's response to COVID-19 was notable for its prolonged school closures, longer than most other countries. Indeed, Chile had the highest number of days with schools fully closed among OECD countries (see Figure 7.1 in the Appendix). Schools initially shut down in March 2020 and began to slowly reopen in August. Yet, by December 2020, only 15% of students had schools with in-person activities to return to them. Then, this rose to 24% in March 2021, 51% in July 2021 and 98% by December 2021. This makes Chile an interesting case study to examine the long-term impacts of such closures and how interrupting (or resuming) inperson classes affected female entrepreneurship by increasing (or reducing) childcare duties.

In Chile, mobility restrictions, public health health measures, and school openness statuses during the pandemic were established and reported weekly at the municipal level. This granularity —covering lockdowns, infection and testing rates, and school statuses—enables our study to differentiate the impact of school closures (and reopenings) from other pandemic-related effects, by exploiting variability at both municipal and individual levels.

4. Methodology

a. Data sources

i. Registry of Companies of Chile (RES)

We use administrative data from the Registry of Companies of Chile (RES), which is the most common way to register a firm nowadays. As Figure 7.2 shows, most firms in Chile are registered through RES. This is mainly because of its relative simplicity and the absence of mandatory physical steps in the whole process. RES contains data of 73.1% of all Chilean registrations of new firms, with their respective date and municipality of registration from May 2013 to December 2021. It is noteworthy that there was a substantial surge in registrations through RES at the end of April 2020. Moreover, as depicted in Figure 7.3 (in the Appendix), the significant impact of the new advanced electronic signature on the establishment of new businesses becomes evident. It is estimated that during its inaugural year of implementation, this reform was responsible for the creation of approximately 30,000 businesses nationwide (Montégu et al., 2021).

In addition to firms' registry information, the RES contains member IDs of registered firms, with some socio demographic information, such as the sex of the entrepreneur. This allows us to identify all people who were involved as a founder in the registration of a new firm and the time and place where it was held. Figure 4.1 shows the weekly evolution of men and women founding partners from January 2019 to December 2021. Here, at first, we see that there are consistently fewer female entrepreneurs than men and that those trends are similar between them, being affected by the same seasonal shocks, at least in the pre-COVID period. Additionally, we also see a sharp decline in both men and women entrepreneurs during the COVID outbreak and mandatory school closures (outlined with the first vertical dashed line). Nonetheless, after the decline in rates during March 2020, there is a rise that coincides with the implementation of the new AES on April 23rd, 2020.



Figure 4.1: Weekly number of entrepreneurs by sex

Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to re-open. Source: Own elaboration, based on public data from RES monthly reports from the Ministry of Economy, Development and Tourism.

Although both groups experienced a decline in March 2020, the extent of this decrease was not uniform. Figure 4.2 depicts the female-to-male entrepreneur ratio, revealing a reduction in the number of women entrepreneurs for every male entrepreneur shortly after the government's mandatory shutdowns were enforced on March 16, 2020. Interestingly, once schools were officially permitted to reopen on August 17th, 2020, the ratio rebounded to levels seen before the COVID-19 pandemic and even exceeded those levels after August 2021.



Figure 4.2: Women to men entrepreneurs' ratio

Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to re-open. Source: Own elaboration, based on public data from RES monthly reports from the Ministry of Economy, Development and Tourism.

ii. Social Household Registry (RSH)

We take advantage of RES data by merging it with the Social Household Registry (Registro Social de Hogares, RSH) from the Ministry of Family and Social Development (Ministerio de Desarrollo Social y Familia, MDSyF).

The RSH is an administrative record that consists of the main instrument for the selection of beneficiaries of subsidies and social programs in the country. It is built with self-reported data from the beneficiaries and government's administrative data coming from several public institutions¹. With this record, we have access to individual information of sociodemographic variables from January 2019 to December 2021. These include people's

¹ For more information, click here.

gender, age, marital status, number of children, ethnicity, labour status, labour participation and labour earnings, among others.

This database also includes information of all the subsidies given by the central government through the Social Information Registry (Registro de Información Social, RIS), from the MDSyF. This is important because, during the pandemic, the Chilean government provided assistance and subsidies totalling US\$38.372 million between 2020 and 2021. Within Latin America, Chile was the country that provided support to families for the longest period (18 months), with significant coverage, reaching 90% of the Social Registry of Households and with an amount provided to families above the poverty line².

iii. Civil registry (SRCeI)

We also include information of the Civil Registry (Servicio de Registro Civil e Identificación, SRCeI). This administrative record contains all parent-child relationships in the national territory, with their respective IDs numbers. This is useful to determine the number of children each person of working age has, along with the respective age of each of them.

iv. COVID-19 data

Additionally, we incorporate public information about the COVID-19 pandemic. In particular, we include data on the implementation of the Step-by-Step Plan (Plan Paso a Paso), the strategic plan designed by the Chilean Ministry of Health (MINSAL) to tackle the coronavirus outbreak, as well as information of epidemiological indicators (e.g. COVID-19 infection and death rates, PCR test rates and test positivity rates) (Pertuzé et al., 2023). This data is made publicly available by the Ministry of Science, Technology, Knowledge and Innovation⁴.

² For more information, click here and here.

⁴ For more information, click here.

The Step-by-Step plan mainly consisted of four different phases, going from the most restrictive (Phase 1 or "Quarantine"), where only essential workers had permission to leave their homes, to the less restrictive (Phase 4 or "Advanced Opening"), were high-contact activities and larger gatherings of people were allowed. Figure 7.4 (in the Appendix) shows the weekly evolution of the Step-by-Step Plan. As we observe, as soon as COVID-19 reached the country, the percentage of the national population under Phase 1 (i.e., full lockdowns) started to rise. Nonetheless, mobility constraints decreased during the spring and summer of 2020, followed by a resurgence in the winter of 2021, reaching its peak during April and May of that year. Epidemiological indicators are illustrated in Figure 7.5 in the Appendix. Here we observe that COVID infection rates reached particularly high levels during the winter months of both 2020 and 2021. Notably, the second wave of infections resulted in fewer fatalities compared to the initial wave.

v. Educational records

We include two datasets from the Ministry of Education (Mineduc). The first one provides school enrolment data spanning the years 2019 to 2021, including the corresponding municipality for each school. These data prove valuable as they facilitate the linkage of children's educational institutions with those of their respective parents, when combined with the RSH and RES datasets. The second dataset from the Mineduc has information on whether each school in the country had in-person classes or not each week during the pandemic. As stated before, all schools in the country were ordered to close by official decree on March 16th, 2020, and were voluntarily allowed to reopen on August 17th of the same year. However, after the central government signed the official decree that allowed them to start doing face-to-face classes again, few schools reopened immediately, and the vast majority took several months to reopen (see Figure 7.6 in the Appendix).

In this vein, schools also tended to react more slowly than phases to the improvement of sanitary conditions. Figure 7.7 in the Appendix shows the evolution of the percentage of students with open schools and phases of the Step-by-Step Plan. As we see, almost all schools were closed during 2020 despite a significant share of the population being in Phase 3 or

Phase 4 (with less mobility restrictions) during that year. Then, only in the second semester of 2021, a larger number of schools started to reopen. This phenomenon could be explained by political pressures, adaptation problems and the requirement to fulfil the sanitary protocols before reopening³.

It is also noteworthy that the rise in the proportion of schools reopening aligns with the increasing trend seen in the ratio of women-to-men entrepreneurs, as illustrated in Figure 4.2. Both series are presented in Figure 4.3. In the pre-COVID era, when it was assumed that all students were attending open schools, the ratio of women to men entrepreneurs fell within the range of 0.5 to 0.55. However, during the period of mandatory school closures, this ratio decreased to approximately 0.45 women for every male entrepreneur. Subsequently, following the government's decision to allow school reopenings, over the next ten months, the ratio rebounded to its pre-COVID levels. Additionally, as the percentage of schools gradually increased towards the end of 2021, the ratio surpassed pre-COVID levels, reaching its peak in October 2021 at 0.65 women for every male entrepreneur.

³ Among the requirements requested by the Ministry of Health to reopen, were the sanitation of the school every 24 hours, the mandatory use of face masks and the distancing of students of at least 1 metre in bathrooms, rooms of classes and common spaces.



Figure 4.3: Women entrepreneurs to men entrepreneurs' ratio and school opening

Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to reopen. Source: Own elaboration, based on public data from RES monthly reports from the Ministry of Economy, Development and Tourism, the Ministry of Science, Technology, Knowledge and Innovation and the Ministry of Education.

To build our final database, we merged all previously mentioned information from the 2019-2021 period on a weekly basis. We also refined our sample to include only working-age men (18-65 years old) and women (18-60 years old) who are parents of school-age children (0-18 years old) and reside in biparental households. As a result, our dataset comprises 107,904,684 fully integrated observations. This constitutes a balanced panel of 781,918 individuals for each week, with 390,959 individuals of each gender. For every observation, we recorded data about whether an individual registered a firm through the RES during a particular week and the municipality in which the individual resided during that time. Additionally, we incorporated the sociodemographic variables from RSH, the school status of their children, and the weekly municipal data from the Step-by-Step Plan and epidemiological controls.

b. Descriptive statistics

Table 1 presents the summary statistics of the entire sample and further breaks it down by gender. The data reveals that both men and women each account for half of the observations. On average, fathers are slightly older (43.1 years) than mothers (40.3 years). Approximately 4% of the sample consists of foreign individuals, and close to 10% identify as members of indigenous peoples.

Regarding employment, an average of 72% of the entire sample is currently employed. This consists of 92% of the male participants and only 53% of the female participants. The average monthly earnings for employed men stands at CLP 459,353, while for women it's CLP 312,432, which is 32% less than their male counterparts. The average household income per individual is approximately CLP 123,533, and the mean socioeconomic quintile is 54.15. Despite earning less, women receive more in state subsidies (averaging CLP 92,906) compared to men (averaging CLP 74,205). During this period, our sample indicates there are, on average, 38.2 male entrepreneurs per 1,000 men and 13.6 female entrepreneurs per 1,000 women.

Considering households with children, both men and women have an average of 1.76 children aged between 3 and 18 years. Breaking this down by age groups, there are on average, 0.31 children aged 3 to 6 years, 0.94 children aged 7 to 14 years, and 0.40 children aged 15 to 18 years. This distribution suggests that the majority of households with children tend to have kids in the middle childhood to early adolescence range (7 to 14 years).

	Full sample Mean (St. dv)	Fathers Mean (St. dv)	Mothers Mean (St. dv)
Demographics			
Age	41.69	43.05	40.33

	(8.65)	(9.12)	(7.92)
Foreign=1	0.04	0.03	0.04
	(0.18)	(0.18)	(0.19)
Ethnicity=1	0.10	0.09	0.11
	(0.3)	(0.29)	(0.31)
Employment and income			
Working=1	0.72	0.92	0.53
	(0.45)	(0.27)	(0.5)
Labour Income (CLP)	\$ 408,501	\$ 459,353	\$ 312,432
	(496366.5)	(552541.88)	(392937.75)
Average income in HH (CLP)	\$ 123,533	\$ 123,533	\$ 123,533
	(146180.3)	(146180.3)	(146180.3)
Income Quintile (from 0 to 100)	54.15	54.15	54.15
	(19.07)	(19.07)	(19.07)
Subsidies (CLP)	\$ 83,563	\$ 74,205	\$ 92,906
	(287221.22)	(255971.2)	(315122.56)
Entrepreneurs per 1,000 individuals	20.6	38.2	13.6
	(0.02)	(0.02)	(0.01)
Children in HH			
N° of Children 3-18	1.76	1.76	1.76
	(0.84)	(0.84)	(0.84)
N° of Children 3-6	0.31	0.31	0.31
	(0.51)	(0.51)	(0.51)
N° of Children 7-14	0.94	0.94	0.94
	(0.71)	(0.71)	(0.71)
N° of Children 15-18	0.40	0.40	0.40
	(0.56)	(0.56)	(0.56)
Observations	107,904,684	53,952,342	53,952,342

Note: The sample includes a balanced panel of fathers and mothers with children aged 3 to 18 years, who lived in biparental households, as indicated by the RSH, throughout the study period (January 2019 to December 2021).

The significant variability observed at both the weekly and municipal levels is critical to our analysis. It captures the differences in school closures, reopenings, and the progression

through the phases of the Step-by-Step Plan—all set against the backdrop of entrepreneurial activity. Figures 7.8 and 7.9 in the Appendix underscore the geographic variation of the Stepby-Step plan and school reopenings, respectively. For several weeks, while certain municipalities grappled with Phase 1 restrictions, others navigated the relative freedom of Phase 4. These disparities are especially evident in school reopening rates during the first half of 2021. This variability enables us to differentiate the effects of the pandemic from those of school closures. In Section 4.c, we leverage this to design an empirical strategy that accurately identifies our coefficients of interest.

c. Empirical Strategy

i. Main regression

In this section, we outline the empirical strategy that allows us to determine the effect of school closures and reopenings on female entrepreneurships. To do this, we estimate the probability of starting a business for each person *i* in municipality *k* during week *t*:

$$Entr_{i,k,t} = \beta_0 + \beta_1 * School_Closure_{i,t} + \beta_2 * School_Reopen_{i,t} + \sum_{j=1}^4 \beta_{j+2} * Phase_{k,t}^j + \beta_7 * COVID_{k,t} + \beta_8 * Woman_{i,t} + \beta_9 * School_Closure_{i,t} * Woman_{i,t} + \beta_{10} * School_Reopen_{i,t} * Woman_{i,t} + \delta_1 * AES_t + \delta_2 * X_{i,t}i + \gamma_k + \mu_t + \epsilon_{k,t}$$

$$(1)$$

Here, $Entr_{i,k,t}$ is a dummy that takes a value of 1 if an individual *i* started a business on week *t* and 0 otherwise. Variable *School_Closure*_{*i*,*t*} is a dummy that takes the value 1 if at least 1 children's schools of person *i* closed during week *t*, while *School_Reopen*_{*i*,*t*} equals 1 when all children's schools opened during week *t*, and 0 otherwise. Therefore, when *School_Closure*_{*i*,*t*} = 1, *School_Reopen*_{*i*,*t*} = 0 and vice versa after March 16th, 2020 (prior to that, both dummies are equal to 0). We further extend this analysis by considering an alternative definition of *School_Closure*_{*i*,*t*} and *School_Reopen*_{*i*,*t*}, where the first variable equals 1 only when all

children's schools of person *i* are closed and the latter is 1 when at least 1 children's school reopened. This extension is showed in Section 6.

Then, $Phase_{i,t}^{j}$ denote 4 dummies that take value 1 when municipality *i* entered phase *j* of the Step-by-Step Plan during week *t*. *Woman*_{*i*,*t*} is a dummy that is equal to 1 when the entrepreneurs are women and 0 otherwise.

Coefficients of interactions between variables *School_Closure*_{*i*,*t*} and *Woman*_{*i*,*t*} and *School_Reopen*_{*i*,*t*}, and *Woman*_{*i*,*t*} are our coefficients of interest, since they show differentiated effects of closure and reopening of schools on the probability of starting a business for fathers and mothers. Finally, *AES*_{*t*} is a dummy that takes value 1 when the Advanced Electronic Signature entered into force in firms' creation system through RES at the national level. *X*_{*i*,*t*} represents controls at the individual level (age, income, number of workers in the household, working status, number of elderly in the household, nationality, ethnicity, subsidies received), while γ_k and μ_t represent fixed effects per municipality and week, respectively. Finally, $\epsilon_{k,t}$ denotes the error term.

ii. Heterogenous effects

We also document potential heterogenous impacts within different sectors of the population. To do this, we disaggregate the sample by 2 different groups: one based on children's ages and the other based on household' socioeconomic quintiles.

For the first disaggregation, we categorize our sample into three distinct groups: parents with children's ages between 3-6 years, 7-14 years and 15-18 years. This categorization aligns with the different school stages in the Chilean educational system, such as pre-school, basic education, and secondary education. Subsequently, for the socioeconomic segmentation, we partition the sample according to the reported data, distinguishing among the following groups: the 40% poorest population (Quintile I and II), Quintile II, Quintile IV, and Quintile V.

5. Results

a. Main Results

Table 2 presents our estimations on how school closures and reopenings affected entrepreneurial activities of fathers and mothers within two-parent households with children aged 3 to 18 years. To delve into potential gender-differentiated effects of school closures on parental employment, our focus remains on heterosexual couples, irrespective of their marital status.

Table 2 display regression coefficients based on the primary definition of school closure and reopening indicators; where $School_Closure_{i,t}$ is set to 1 if at least one child's school is closed, and $School_Reopen_{i,t}$ is set to 1 when all children's schools have reopened. Additionally, in Section 6, we consider alternative definitions of this variable employing a more nuanced definition for these indicators; here, $School_Closure_{i,t}$ is set to 1 when all children's schools are closed, and $School_Reopen_{i,t}$ is set to 1 if at least one child's school has reopened. Furthermore, column (1) of Table 2 presents a model without fixed effects or COVID-specific controls. In contrast, column (2), detail a model that incorporates sociodemographic controls and fixed effects (FE), while column (3) represents the full model (with sociodemographic controls, fixed effects and COVID controls).

Table 2 indicates that school closures led to a notable decrease in entrepreneurship during the pandemic. As depicted in column (3), school closures reduced the likelihood of initiating a business by -0.01 percentage points (p.p.) each week. Given the size of our sample, we deduce that school closures deterred on average 82 parents of children aged 3-18 from registering a company through RES weekly. More specifically, this corresponds to 41 men and 41 women.

The interaction between the closure variable and the gender dummy suggests a more pronounced decline in entrepreneurship among mothers compared to fathers. We estimate

that school closures diminished the weekly probability of women starting a business by an additional -0.005 p.p. Considering the size of our panel, our estimates indicate that school closures barred an additional 18 mothers from launching a business each week. This translates to a cumulative negative impact of 59 weekly registrations. Furthermore, considering the gender gap before the school closures initiated (captured by the variable *Woman*), we can conclude that school closures widened the gender gap by 29,6%⁴.

We now explore whether the dynamics shifted when schools started reopening. Interestingly, we cannot reject the null hypothesis that, upon school reopenings, the overall likelihood of starting a business equaled that of pre-closure periods. The coefficient of column (3) is negative but not statistically significant at the 5% level. However, the likelihood of starting a business for mothers with children aged 3-18 after the reopen of schools decreased by -0.004 p.p. each week. Applying this estimate to our sample implies that roughly 14 mothers were dissuaded each week from starting their businesses. While there was a modest increase when schools reopened, it did not suffice to return the entrepreneurship rates of mothers to pre-pandemic levels. Furthermore, considering again the gender gap before the school closures initiated (captured by the variable *Woman*), we can conclude that school reopened increased the gender gap by "only" 22,9%⁵.

These results suggests that women entrepreneurs sharply declined upon school closure and increased upon school reopening, but that this increase upon reopening was not sufficient to recover pre-school closure baseline rates. This can be explained by the low attendance of students during the beginning of reopen. In fact, the return of children to school occurred significantly later following the reopening. For example, attendance remained voluntary up to March 2022, with only between 35% and 55% of students attending school each day by the end of the academic year of 2021 (Claro et al., 2021).

⁴ This is calculated by dividing coefficients of *School_closure#Woman* and *Woman:* -0.0000468/-0.000158=29.6%.

⁵ As before, this is calculated by dividing the coefficients of *School_reopen#Woman* and *Woman:* -0.0000363/-0.000158=22,9%.

The uneven responses between men and women, as evidenced in Table 2, resonate with existing literature on parental time allocation. Various studies emphasise that during the early phases of the pandemic, mothers predominantly assumed the bulk of childcare responsibilities. (Amuedo-Dorantes, et al., 2023; Adams-Prassl, et al., 2020; Alon et al., 2020, 2021).

		Main Regression	
	(1)	(2)	(3)
	Sociodemographic controls	Sociodemographic controls & FE	Full model
School closure	-0.000500***	-0.0000770***	-0.000104**
	(-6.63)	(-8.82)	(-3.26)
Woman	-0.000157***	-0.000158***	-0.000158***
	(-32.93)	(-33.13)	(-27.12)
School closure#Mother	-0.0000468***	-0.0000469***	-0.0000468***
	(-7.44)	(-7.44)	(-7.39)
School reopen	0.00000333	0.0000329**	-0.0000626
	(0.35)	(-3.17)	(-1.87)
School reopen#Mother	-0.0000364***	-0.0000362***	-0.0000363***
	(-4.03)	(-4.01)	(-4.27)
AES	0.000151***	0.000178***	0.000156***
	(21.71)	(21.71)	(16.98)
Age	-0.0000864***	-0.00000867***	-0.00000869***
	(-47.64)	(-47.67)	(-24.92)
Income (CLP)	6.37e-10***	5.97e-10****	5.97e-10***
	(61.41)	(56.52)	(19.76)
N° of workers in HH	-0.0000321	-0.00000544*	-0.00000522
	(-1.17)	(-1.98)	(-1.48)
Working (RSH)	0.0000966*	0.00000714	0.00000718
	(-2.15)	(1.58)	(1.57)
N° of elderly in HH	0.0000805***	0.0000683***	0.0000684***
,	(7.13)	(6.04)	(3.83)

Table 2: Entrepreneurial response to school closures and reopen of fathers and mothers ofbiparental households.

Foreign	0.000131*** (16.36)	0.0000978*** (11.41)	0.0000977*** (4.61)
Ethnicity	-0.000433***	-0.0000396***	-0.0000396***
	(-8.88)	(-7.28)	(-5.90)
Subsidies (CLP)	2.21e-12	4.51e-12	3.22e-12
	(0.42)	(0.85)	(0.56)
Female pre-closure baseline mean	0.0001231	0.0001231	0.0001231
Municipal & Week FE	No	Yes	Yes
Phases & epidemiological controls	No	No	Yes
Observations	107,904,684	107,904,684	107,904,684

t statistics in parentheses

*p<0.05, ** p<0.01, *** p<0.001,

b. Heterogeneous impacts

In this section, we document the heterogeneous effects of school closures and reopenings. At first, we show the results on fathers and mothers with children of various ages, enabling us to discern the specific contexts in which this impact is most pronounced. For this purpose, we categorise parents into three distinct groups based on the presence of at least one child in specific age brackets: 3-6 years (preschoolers), 7-14 years (elementary and middle school), and 15-18 years (high school).

i. By children's age

Table 3 presents variations in estimates for each parental group. As in column (3) of Table 2, coefficients are based on models with all controls and fixed effects. Here, the interaction estimates of school closures and reopenings with the gender dummy vary between groups. For instance, school closures had a considerably smaller impact among women with children aged 15 to 18 years, leading to a decline in the probability of starting a business by "only" -0.00239 p.p. each week. In contrast, women with children aged 3 to 6 years faced the most significant effects from school closures, with a drop of -0.00734 p.p. in their probability of starting a business each week. Thus, women in the second group were around three times more affected than those in the first. Mothers with children 7-14 years experienced the second most pronounced effects, with a decrease in likelihood of -0.00529 p.p.

Moreover, the impact on the gender gap varied across different groups. Assessing the gender gap during the pre-closure period (reflected by the *Woman* coefficient), school closures notably amplified this gap, particularly among mothers with children aged 3 to 6 years, resulting in a 41.9% widening. In contrast, this gap increased by 31.9% among parents with children aged 7 to 14 years and by a comparatively lower 14.4% among parents with children aged 15 to 18 years.

Then, for two out of the three groups, we cannot dismiss the possibility that, upon school reopenings, entrepreneurial trends resembled those of pre-closure periods, even though coefficients for all age groups were negative. These results were not statistically different from zero at a 5% level. The only group exhibiting a significant decline comprised parents with children aged 3-6 years. Lastly, when examining the interaction between school reopenings and the gender dummy, we noted that the effect on mothers with children aged 15-18 years wasn't statistically significant. In contrast, the most pronounced effect occurred again among women with children aged 7-14 years. Here, the gender gap increased by "only" 27.9% comparing with the pre-closure period, a similar result for the group with children aged 3-6 years (25.5%). By contrast, the coefficient on the group with children aged 15-18 years was negative but not significant at a 5% confidence level.

Table 3: Entrepreneurial response to school closures and reopen of biparental households with children ages 3–18.

		Parents with	
	Parents with children between	children between 7-14	Parents with children between
	3-6 years	years	15-18 years
School closure	-0.000145**	-0.000101***	-0.0000958*
	(-3.17)	(-3.82)	(-2.55)
Woman	-0.000175***	-0.000166***	-0.000166***
School closure#Mother	(-18.97) -0.0000734*** (-5.74)	(-30.79) -0.0000529*** (-7.29)	(-21.03) -0.0000239* (-2.44)

School reopen	-0.000103*	-0.0000513	-0.0000475
School reopen#Mother	(-2.10) -0.0000447* (-2.05)	(-1.82) -0.0000464*** (-3.93)	(-1.21) -0.0000267 (-1.89)
Female pre- closure baseline mean	0.0001476	0.0001239	0.001169
Sociodemographic & AES controls	Yes	Yes	Yes
Municipal & Week FE	Yes	Yes	Yes
Phases & epidemiological controls	Yes	Yes	Yes
Observations	30,370,417	79,508,785	39,132,524

t statistics in parentheses.

*p<0.05, ** p<0.01, *** p<0.001.

ii. By socioeconomic quintiles

We also document results differentiating by socioeconomic quintiles. This classification divides the Chilean population by income levels. This is calculated by taking the overall household income and dividing it by the number of family members. These categories are already calculated in the RSH database, where quintiles I and II (representing lower-income groups) are consolidated into a single category.

Table 4 presents variations in estimates for each socioeconomic group. As in Table 3, coefficients are based on models with all controls and fixed effects. Here, the interaction estimates of school closures and reopenings with the gender dummy vary between groups. For instance, school closures had a considerably higher impact among women from the highest quintile income, leading to a decline in the probability of starting a business by -0.00691 p.p. each week. By contrast, women from the lowest quintiles (I, II and III) faced the lowest impact (-0.00454 p.p. and -0.00428 p.p., respectively).

Nevertheless, there was a significant difference in the probability of starting a business for women in each group during the pre-closure period. While quintile V had a probability of starting a business of 0.000313 each week during that period, for quintile I and II this was only 0.0000781, meaning that women in the first group were 4 times more likely to start a business before school closures. Thus, coefficients of interactions should be interpreted taking this into account. Then, if we consider the relative effect of school closures with

respect to the pre-closure female mean, the most affected group are women from quintile I and II, reaching a decrease of 58,1%⁶ of starting a business. On the other side, quintile III suffered a 40% decreased, while the relative change for quintile IV was 35,9%. Finally, quintile V was the least affected with a relative change of "only" 22,1%.

Furthermore, considering the gender gap before the school closures initiated (captured by the variable *Woman*), we can conclude that school closures widened the gender gap by 38,8%, 28,5%, 28,8% and 21,6% for quintile I-II, III, IV and V, respectively.

Then, for two out of the four groups, we cannot dismiss the possibility that, upon school reopenings, entrepreneurial trends resembled those of pre-closure periods, even though coefficients for all groups were negative. The only group exhibiting a significant decline comprised parents from quintile I and II. Lastly, when examining the interaction between school reopenings and the gender dummy, we noted that the effect on mothers from quintiles IV and V were not statistically significant. In contrast, the most pronounced relative effect occurred among women from quintile III. Here, the relative change of quintiles I-II, III, IV and V were 45,3%, 47,1%, 12,1% and 10%, respectively.

Finally, the gender gap increased by "only" 30,3% comparing with the pre-closure period for quintiles I-II, a similar result for the group from quintile III (33,6%). By contrast, the coefficients on quintiles IV and V were negative but not significant at a 5% confidence level.

Table 4: Entrepreneurial response to school closures and reopen of biparental householdsby socioeconomic quintiles.

	Quintile I and II	Quintile III	Quintile IV	Quintile V
School closure	-0.0000987**	-0.000112*	-0.000610	-0.000157
	(-3.02)	(-2.10)	(-0.87)	(-1.55)
Woman	-0.000117***	-0.000150***	-0.000182***	-0.000319***
	(-18.74)	(-14.83)	(-14.43)	(-17.64)

⁶-0,0000454/0,00007809=58,1%.

School closure#Mother	-0.0000454***	-0.0000428**	-0.0000525***	-0.0000691*
	(-6.39)	(-3.06)	(-3.68)	(-2.55)
School reopen	-0.0000478	-0.000371	-0.0000666	-0.000185
	(-1.34)	(-0.66)	(-0.89)	(-1.76)
School reopen#Mother	-0.0000354***	-0.0000504*	-0.0000176	-0.0000312
	(-3.37)	(-2.50)	(-0.79)	(-0.83)
Female pre-closure baseline mean	0.00007809	0.000107	0.000146	0.000313
Sociodemographic & AES controls	Yes	Yes	Yes	Yes
Municipal & Week FE	Yes	Yes	Yes	Yes
Phases & epidemiological controls	Yes	Yes	Yes	Yes
Observations	58,466,747	19,120,474	15,312,239	13,236,202

t statistics in parentheses

*p<0.05, ** p<0.01, *** p<0.001,

6. Robustness checks

Additionally, in Section 6, we consider alternative definitions of our variables of interest, employing a more nuanced definition for these indicators; here, $School_Closure_{i,t}$ is set to 1 when all children's schools are closed, and $School_Reopen_{i,t}$ is set to 1 if at least one child's school has reopened.

Columns (4), (5), and (6) of Table 5 present results using an alternative definition for school closures and reopenings. Notably, the outcomes align closely with those in columns (1) through (3) of Table 2. Under this definition, school closures continue to exert a significant, and nearly identical, influence on overall entrepreneurship rates. However, the impact on female entrepreneurship is marginally reduced, which is congruent with the less stringent definition where the variable is set to 1 only when all children's schools are closed. In addition, school reopenings also reflect a negative but not significant effect at the 5% level on overall entrepreneurship in column (6). Conversely, the interaction with the gender variable indicates a marginally greater negative influence on the likelihood of mothers initiating businesses compared to the results in column (3) of Table 2.

Furthermore, alternative models for differentiated effects on children's age and socioeconomic quintiles (in Appendix b) present again similar results to those in Table 3 and 4, giving confidence about the main results

Table 5: Entrepreneurial response to school closures and reopen of fathers and mothersof biparental households.

	School closures & reopen 2			
	(4)	(5)	(6)	
	Only controls	FE	FE & controls	
School closure	-0.000509***	-0.0000785***	-0.000101***	
	(-6.82)	(-9.08)	(-4.55)	
Woman	-0.000156***	-0.000156***	-0.000157***	
	(-33.10)	(-33.29)	(-33.29)	
School closure#Mother	-0.0000464***	-0.0000464***	-0.0000464***	
	(-7.39)	(-7.40)	(-7.39)	
School reopen	0.0000239	0.0000343	-0.0000579*	
-	(0.26)	(-3.38)	(-2.53)	
School reopen#Mother	-0.0000386***	-0.0000385***	-0.0000385***	
-	(-4.4)	(-4.42)	(-4.43)	
AES	0.00015=***	0.000178***	0.000156***	
	(21.82)	(22.01)	-17.01	
Age	-0.00000862***	-0.00000863***	-0.00000865***	
0	(-49.35)	(-49.27)	(-49.35)	
Income (CLP)	6.32e-10***	-5.93e-10****	5.93e-10***	
	(61.60)	(56.70)	(56.70)	
N° of workers in HH	-0.0000351	-0.000000591	-0.00000570*	
	(-1.30)	(-2.17)	(-2.10)	
Working (RSH)	0.0000125**	0.00012**	0.0000101*	
	(-2.81)	(2.25)	(2.27)	
N° of elderly in HH	0.0000619***	0.0000542***	0.0000544***	
5	(6.83)	(5.97)	(5.99)	
Foreign	0.000132***	0.000985***	0.0000984***	
	(16.68)	(11.65)	(11.65)	
Ethnicity	-0.000432***	-0.0000398***	-0.0000398***	
2	(-8.94)	(-7.38)	(-7.39)	
Subsidies (CLP)	1.41e-12	3.71e-12	2.49e-12	
	(0.27)	(0.71)	(0.48)	

Female pre-closure baseline mean	0.0001231	0.0001231	0.0001231
Municipal & Week FE	No	Yes	Yes
Phases & epidemiological controls	No	No	Yes
Observations	107,904,684	107,904,684	107,904,684

t statistics in parentheses

*p<0.05, ** p<0.01, *** p<0.001,

7. Discussion

In this research, we evaluate the effects of pandemic-induced school closures on female entrepreneurship in Chile. Our results, based on the analysis of administrative microdata from multiple public agencies, highlight the critical role of institutions in shaping entrepreneurial decisions and outcomes.

Schools, in addition to their primary educational role, serve as an essential support system for working parents. When schools closed due to the pandemic, this support was suddenly withdrawn, causing significant disruption. Our data reveals that this had a particularly pronounced effect on mothers, who often bear a larger share of childcare responsibilities. This observation aligns with past research that has spotlighted the gendered challenges in securing funding, balancing work-life dynamics, and navigating the socio-cultural expectations unique to women entrepreneurs (Jennings and Brush, 2013; Thébaud, 2010, 2015; Castellaneta et al., 2020). "Intensive mothering", a concept reflecting the expectation that mothers should always be available and deeply involved in their children's activities, along with the societal norm that assigns household chores primarily to women (Hochschild and Machung, 2012), reduces mothers' time and resources to start and operate a new venture. This, in turn, exacerbates gender disparities in the rate of new business creation and pushes women towards less growth-oriented forms of entrepreneurship to mitigate workfamily conflicts (Thébaud, 2015). Another notable finding of our study is the varied impact of school closures based on children's ages. Mothers with children between 3-6 years experienced a more significant decline in their entrepreneurial activities compared to those with older children. Furthermore, mothers from poorer backgrounds (quintiles I and II) are also the most affected group, compared with women with higher incomes. These insights are critical for policymakers as they point to specific population groups where support mechanisms might be most needed.

Chile, despite its advancements in various spheres, still contends with deeply ingrained informal institutions—societal expectations and cultural norms—that disproportionately burden women with caregiving responsibilities. For academics and policymakers, our study underscores the importance of institutions, both formal (e.g. schools) and informal (e.g. societal norms), in influencing entrepreneurial activities. While the negative effects of school closures on female entrepreneurship exposed an unintended consequence of such measures, they also revealed deeper, longstanding barriers that women face in their entrepreneurial journeys. Understanding the broader implications of the COVID-19 induced school closures provides an opportunity to create a more supportive environment for female entrepreneurs in the country. This will require targeted policies and a broader shift in societal perceptions, ensuring that the entrepreneurial field becomes more equitable and resilient.

8. Conclusions

In this study, we evaluated the consequences of COVID-19 school closures on entrepreneurial activity in Chile, revealing a deeper manifestation of gender disparities. Our findings shed light on an area of interest that has recently gained traction among academics, who have emphasised the need to understand the ramifications of pandemic-related school closures.

Chile's business and entrepreneurial landscape, much like the rest of the world, faced unprecedented challenges during the pandemic. Yet, it was the intersection of this global crisis with local factors, such as the country's specific pandemic responses, that created a unique context deserving of in-depth analysis. Our findings demonstrate the critical function schools play, not merely as educational institutions, but as pillars upholding women's labour and entrepreneurial activities.

Given the centrality of schools in Chilean society and their role in facilitating a balanced worklife environment, their prolonged closure brought into sharp focus the additional pressures women entrepreneurs had to contend with. These pressures were not just about managing businesses, but also the intensified domestic responsibilities that came in the absence of educational institutions. The extended school closures and the resultant impact on women's entrepreneurial ventures serve as a clear indication that even formal institutions like schools have profound, indirect effects on the broader economic fabric of society.

While our research has provided empirically grounded evidence on some aspects of this complex issue, several areas remain open for further investigation. For instance, the entrepreneurial challenges might differ based on the nature of businesses. Are tech-based startups, which might offer more flexible working hours, less affected than traditional enterprises? Additionally, could there be variations in impact based on differing education levels or household income among women? Such granular analyses could offer further nuances to our understanding.

One limitation of our study is our reliance on administrative records. While comprehensive, they do not provide a window into the lived experiences of these women entrepreneurs. Detailed surveys or qualitative studies could illuminate the daily challenges these women grappled with, from work-family conflicts to the reshuffling of priorities in the face of prolonged school closures.

Furthermore, while Chile provides a compelling case study due to its unique pandemic response trajectory, the findings might vary in different national contexts. Countries with shorter lockdowns, alternate educational continuity measures, or different economic safety nets might present different results. It would be important for future research to conduct comparative studies across different countries, considering their distinct institutional and socio-economic environments.

In wrapping up, our study underscores the multiple challenges that female entrepreneurs in Chile faced during the pandemic. As countries navigate the aftermath of the pandemic, insights such as ours will be necessary for guiding not only academics but also policymakers as they design measures to support entrepreneurs, ensuring that economies remain resilient and inclusive. In Chile, as in many parts of the world, women entrepreneurs are critical to the economy's vitality. Ensuring their challenges are understood and addressed is not just an academic endeavour, but an economic and social imperative.

9. References

Acs, Z.J., & Szerb, L. (2007). Entrepreneurship, Economic Growth and Public Policy. *Small Business Economics*, 28, 109–122.

Adams-Prassl, A., Boneva, T., Golin, M., & Rauh, C. (2020). Inequality in the impact of the coronavirus shock: Evidence from real time surveys. *Journal of Public Economics*, 189.

Agier, I., & Szafarz, A. (2013). "Microfinance and Gender: Is There a Glass Ceiling on Loan Size?," *World Development*, 42, 165-181.

Agostinelli, F., Doepke, M., Sorrenti, G., & Zilibotti, F. (2022). When the great equalizer shuts down: Schools, peers, and parents in pandemic times. *Journal of Public Economics*, 206.

Aguilera, X., Mundt, A. P., Araos, R., & Weitzel, T. (2021). The story behind Chile's rapid rollout of COVID-19 vaccination. *Travel Medicine and Infectious Disease*, 42, 102092.

Ahl, H., & Marlow, S. (2012). Exploring the dynamics of gender, feminism and entrepreneurship: advancing debate to escape a dead end?. *Organization*, 19(5), 543–562.

Albanesi, S. & Kim, J. (2021). Effects of the COVID-19 recession on the US labor market: Occupation, family, and gender. *Journal of Economic Perspectives*, 35(3), 3-24.

Allen, W. D. (2000). Social networks and self-employment. *The Journal of Socioeconomics*, 29(5), 487-501.

Alon, T., Doepke, M., Olmstead-Rumsey, J., & Tertilt, M. (2020). *The impact of COVID-19 on gender equality* (Working Paper No. 26947). National Bureau of Economic Research (NBER).

Alon, T., Coskun, S., Doepke, M., Koll, D., & Tertilt, M. (2021). *From mancession to shecession: Women's employment in regular and pandemic recession* (Working Paper No. 28632). National Bureau of Economic Research (NBER).

Amuedo-Dorantes, C., Marcén, M., Morales, M., & Sevilla, A. (2023). Schooling and parental labor supply: Evidence from COVID-19 school closures in the United States. *ILR Review*, 76(1), p. 68-69.

Audretsch, D. B. (2007). The Entrepreneurial Society. New York: Oxford University Press.

Audretsch, D. B. & Petra Moog (2022). Democracy and Entrepreneurship. *Entrepreneurship Theory and Practice*, 46(2), 368–392.

Autio, E., Pathak, S. & Wennberg, K. (2013). Consequences of cultural practices for entrepreneurial behaviors. *Journal of International Business Studies*, 44, 334–362.

Baumol, W. J. (1990). Entrepreneurship: Productive, Unproductive, and Destructive. *Journal of Political Economy*, 98(5), 893–921.

Beck, T., Demirgüç-Kunt, A., & Maksimovic, V. (2008). Financing patterns around the world: Are small firms different?. *Journal of Financial Economics*, 89(3), 467–487.

Belitski, M., Guenther, C., Kritikos, A.S. et al. (2022). Economic effects of the COVID-19 pandemic on entrepreneurship and small businesses. *Small Business Economics*, 58, 593–609.

Bertrand, M. (2018). Coase Lecture - The Glass Ceiling. *Economica*, 85(338), 205–231.

Betthäuser, B.A., Bach-Mortensen, A.M. & Engzell, P. A systematic review and meta-analysis of the evidence on learning during the COVID-19 pandemic. *Nature Human Behavior* 7, 375–385 (2023).
Black, S. E., & Strahan, P.E. (2002). Entrepreneurship and Bank Credit Availability. The *Journal* of *Finance*, 57(6), 2807–2833.

Boden Jr, R. J., & Nucci, A. R. (2000). On the survival prospects of men's and women's new business ventures. *Journal of Business Venturing*, 15(4), 347-362.

Botero, J. C., Djankov, S., La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2004). The Regulation of Labor. *The Quarterly Journal of Economics*, 119(4), 1339–1382.

Boudreaux, C. J., Nikolaev, B. N., & Klein, P. (2019). Socio-cognitive traits and entrepreneurship: The moderating role of economic institutions. *Journal of Business Venturing*, 34(1), 178-196.

Brana, S. (2013). Microcredit: an answer to the gender problem in funding?. *Small Business Economics*, 40, 87–100.

Brown, R., & Rocha, A. (2020). Entrepreneurial uncertainty during the Covid-19 crisis: Mapping the temporal dynamics of entrepreneurial finance. *Journal of Business Venturing Insights*, 14.

Brush, C., Edelman, L.F., Manolova, T. et al. (2019). A gendered look at entrepreneurship ecosystems. *Small Business Economics*, 53, 393–408.

Brush, C.G., Greene, P.G. (2021). Do Women Engage Differently in Entrepreneurship?. In: Cooney, T.M. (eds) *The Palgrave Handbook of Minority Entrepreneurship.* Palgrave Macmillan, Cham.

Bruton, G. D., Ahlstrom, D., & Li, H. (2010). Institutional Theory and Entrepreneurship: Where Are We Now and Where Do We Need to Move in the Future? *Entrepreneurship Theory and Practice*, 34(3), 421–440.

Busenitz, L. W., Gómez, C., & Spencer, J. W. (2000). Country Institutional Profiles: Unlocking Entrepreneurial Phenomena. *The Academy of Management Journal*, 43(5), 994–1003.

Castellaneta, F., Conti, R., & Kacperczyk, A. (2020). The (un) intended consequences of institutions lowering barriers to entrepreneurship: The impact on female workers. *Strategic Management Journal*, 41(7), 1274-1304.

Clarke, D., Larroulet, P., Pailañir, D., & Quintana, D. (2022). *Schools as a Safety-net: The Impact of School Closures and Reopenings on Rates of Reporting of Violence Against Children*. arXiv.

Claro, S., J. P. Valenzuela, E. A. Undurraga, D. Kuzmanic, and D. Cerda (2021). *Encuesta para monitoreo de colegios abiertos en tiempos de pandemia, pp. 69–114.* Pontificia Universidad Catolica de Chile.

Coleman, S. (2007). The role of human and financial capital in the profitability and growth of women-owned small firms, *Journal of Small Business Management*, 45(3), 303–319.

Coleman, S., & Robb, A. (2009). A comparison of new firm financing by gender: evidence from the Kauffman Firm Survey data, *Small Business Economics*, 33, 397–411,

Cordier, I., & Bade, M. (2023). The relationship between business regulation and nascent and young business entrepreneurship revisited. *Small Business Economics*, 61, 587–616.

Craig, L., Powell, A., & Cortis, N. (2012). Self-employment, work-family time and the gender division of labour. Work, Employment and Society, 26(5), 716–734.

Dasgupta, S. (2022). School in the time of Covid. *Monash Bioeth*. Rev. 40, 120–144.

Davidsson, P., Henrekson, M. (2002). Determinants of the Prevalence of Start-ups and High-Growth Firms. *Small Business Economics*, 19, 81–104.

Deryugina, T., Shurchkov, O., & Stearns, J. E. (2021). *COVID-19 Disruptions Disproportionately Affect Female Academics* (Working Paper No. 28360). National Bureau of Economic Research.

De Soto, H. (1990). *The Other Path.* New York, NY: Harper and Row.

Del Boca, D., Oggero, N., Profeta, P., & Rossi, M. (2020). Women's and men's work, housework and childcare, before and during COVID-19. *Review of Economics of the Household*, 18, 1001–1017.

Desai, M., Gompers, P., & Lerner, J. (2003). *Institutions, Capital Constraints and Entrepreneurial Firm Dynamics: Evidence from Europe* (No. w10165). National Bureau of Economic Research.

Djankov, S., La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2002). The regulation of entry. *Quarterly Journal of Economics*, 117(1), 1-37.

Djankov, S. (2009). The Regulation of Entry: A Survey. *The World Bank Research Observer*, 24(2), 183–203.

Eddleston, K. A., Ladge, J. J., Mitteness, C., & Balachandra, L. (2016). Do you See what I See? Signaling Effects of Gender and Firm Characteristics on Financing Entrepreneurial Ventures. *Entrepreneurship Theory and Practice*, 40(3), 489–514.

Edwards, L. N., & Field-Hendrey, E. (2002). Home-based work and women's labor force decisions. *Journal of Labor Economics*, 20(1), 170–200.

Essers, D., Megersa, K., & Sanfilippo, M. (2021). The Productivity Gaps of Female-Owned Firms: Evidence from Ethiopian Census Data. *Economic Development and Cultural Change*, 69(2), 645-683.

Estrin, S., Korosteleva, J., & Mickiewicz, T. (2013). Which institutions encourage entrepreneurial growth aspirations? *Journal of Business Venturing*, 28(4), 564-580.

Hale, T., Angrist, N., Goldszmidt, R. et al. (2021). A global panel database of pandemic policies (Oxford COVID-19 Government Response Tracker). *Nature Human Behaviour*, 5, 529–538.

Hanushek, E. A., & Woessmann, L. (2020). *The Economic Impacts of Learning Losses. Education Working Papers*, No. 225, OECD Publishing, Paris.

Fairlie R. (2020). The impact of COVID-19 on small business owners: Evidence from the first three months after widespread social-distancing restrictions. *J Econ Manag Strategy*. Winter;29(4):727-740.

Fairlie, R., Fossen, F.M., Johnsen, R. et al. (2023). Were small businesses more likely to permanently close in the pandemic?. *Small Business Economics*, 60, 1613–1629.

Fairlie, R. W., & Robb, A. M. (2009). Gender differences in business performance: evidence from the characteristics of business owners survey. *Small Business Economics*, 33, 375–395.

Farré, L., Fawaz, Y., González, L., & Graves, J. (2020). *How the COVID-19 lockdown affected gender inequality in paid and unpaid work in Spain (IZA Discussion Paper* No. 13434). IZA – Institute of Labor Economics.

Fortin, N. M. (2005). Gender Role Attitudes and the Labour-Market Outcomes of Women Across OECD Countries. *Oxford Review of Economic Policy*, 21(3), 416–438.

Foss, L., Henry, C., Ahl, H. et al. (2019). Women's entrepreneurship policy research: a 30-year review of the evidence. *Small Business Economics*, 53, 409–429.

Gompers, P., & Lerner, J. (1999). The Venture Capital Cycle. MIT Press

Gouëdard, P., B. Pont and R. Viennet (2020), "*Education responses to COVID-19: shaping an implementation strategy*", OECD Education Working Papers, No. 224.

Graeber, D., Kritikos, A.S. & Seebauer, J. (2021). COVID-19: A crisis of the female self-employed. *Journal of Population Economics*, 34, 1141–1187.

Hardy, M, and Kagy, G. (2018). "Mind The (Profit) Gap: Why Are Female Enterprise Owners Earning Less Than Men?" *AEA Papers and Proceedings*, 108, 252-255.

Harris, D. N., Ziedan, E., & Hassig, S. (2021). *The Effects of School Reopenings on COVID-19 Hospitalizations.* National Center for Research on Education Access and Choice (REACH). Technical Report.

Hayton, J.C., Gabriella, C. (2013). "Is there an entrepreneurial culture? A review of empirical research". Entrepreneurship & Regional Development, 25(9-10), 708–731.

Hipp, L. & Bünning, M. (2021). *Parenthood as a driver of increased gender inequality during COVID-19? Exploratory evidence from Germany.* European Societies, 23(1).

Hochschild, A., & Machung, A. (2012). The second shift: Working families and the revolution at *home.* Penguin.

Hyytinen, A., & Ruuskanen, O.P. (2007). *Time Use of the Self-Employed*. Kyklos, 60(1), 105–122.

Jennings, J.E., Brush, C.G. (2013). Research on Women Entrepreneurs: Challenges to (and from) the Broader Entrepreneurship Literature?. *Academy of Management Annals*, 7(1), 663–715.

Kenny, C., & Yang, G. (2021). *The Global Childcare Workload from School and Preschool Closures During the COVID-19 Pandemic.* Center for Global Development Note (June 2021).

Kiefer, K., Heileman, M. & Pett, T.L. (2022). Does gender still matter? An examination of small business performance. *Small Bus Econ* 58, 141–167-

Kuhfeld, M., Soland, J., & Lewis, K. (2022). Test score patterns across three COVID-19-impacted school years. *Educational Researcher*, 51(7), 500–506.

Kuzmanic, D., Valenzuela, J. P., Claro, S., Canales, A., Cerda, D., & Undurraga, E. A. (2023). Socioeconomic disparities in the reopening of schools during the pandemic in Chile. *International Journal of Educational Development*, 100.

Klapper, L., Laeven, L., & Rajan, R. (2006). Entry regulation as a barrier to entrepreneurship. *Journal of Financial Economics*, 82(3), 591–629

Laffont, J. J., & Tirole, J. (2001). *Competition in Telecommunications*. MIT Press.

Lauto, G., Salvador, E. et Visintin, F. (2022). "For what they are, not for what they bring: The signaling value of gender for financial resource acquisition in academic spin-offs". *Research Policy*, 51(7).

Liu, Y., Wei, S., Xu, J. (2021). *COVID-19 and women-led businesses around the world.* Finance Research Letters, 43.

Lombard, K. V. (2001). Female self-employment and demand for flexible, nonstandard work schedules, *Economic Inquiry*, 39, pp. 214–37.

Loscocco, K., & Bird, S. R. (2012). Gendered paths: Why women lag behind men in small business success. *Work and Occupations*, 39(2), 183–219.

Lusardi, A., & Mitchell, O. (2011). Financial literacy around the world: An overview. *Journal of Pension Economics & Finance*, 10(4), 497-508.

Mas, A., & Pallais, A. (2017). Valuing alternative work arrangements. *American Economic Review*, 107(12), 3722–59.

McAdam, M., Harrison, R.T. & Leitch, C.M. (2019). Stories from the field: women's networking as gender capital in entrepreneurial ecosystems. *Small Business Economics*, 53, 459–474.

McMullen, J. S., Bagby, D. R., & Palich, L. E. (2008). Economic Freedom and the Motivation to Engage in Entrepreneurial Action. *Entrepreneurship Theory and Practice*, 32(5), 875–895.

Minniti, M. (2008). The role of government policy on entrepreneurial activity: productive, unproductive, or destructive?. *Entrepreneurship Theory and Practice*, 32(5), 779-790.

Montégu, J.P., Quiroga, J.T., & Pertuze, J. A. (2021). *Institutional responses for promoting entrepreneurship: Red tape reduction during the COVID-19 crisis*. Working paper.

Myers, K.R., Tham, W.Y., Yin, Y. et al. (2020). Unequal effects of the COVID-19 pandemic on scientists. *Nature Human Behaviour*, 4, 880–883.

Neumeyer, X., Santos, S.C., Caetano, A. et al. (2019). Entrepreneurship ecosystems and women entrepreneurs: a social capital and network approach. *Small Bus Econ* 53, 475–489.

North, D. C. (1990). *Institutions, institutional change and economic performance.* Cambridge University Press.

OECD (2017), G20/OECD INFE report on adult financial literacy in G20 countries.

OECD. (2022). Education at a Glance 2022: OECD Indicators, OECD Publishing, Paris.

OECD/EU. (2017). *Policy brief on women's entrepreneurship,* OECD Publishing. <u>https://data.europa.eu/doi/10.2767/50209</u>

Pearson, H. (2022). COVID derailed learning for 1.6 billion students. Here's how schools can help them catch up. *Nature*, 605, 608-611. <u>https://doi.org/10.1038/d41586-022-01387-7</u>

Pertuze, J. A., Montégu, J. P., González, C., Araos, R., & Daza, P. (2023). Navigating economic turmoil: Chilean businesses during COVID-19 lockdowns and vaccine rollouts. *Health Policy and Technology*, 100813. <u>https://doi.org/10.1016/j.hlpt.2023.100813</u>

Ratten, V. (2020). Coronavirus (covid 19) and entrepreneurship: changing life and work landscape. *Journal of Small Business and Entrepreneurship*, Vol. 32 No. 5, pp. 503-516.

Renzulli, L. A., Aldrich, H., & Moody, J. (2000). Family Matters: Gender, Networks, and Entrepreneurial Outcomes. *Social Forces*, 79(2), 523–546.

Robb, A.M., Robinson, D.T. (2014). The Capital Structure Decisions of New Firms. *Review of Financial Studies*, 27(1), 153–179.

Schutjens, V., & Stam, E. (2003). The Evolution and Nature of Young Firm Networks: A longitudinal perspective. *Small Business Economics*, 21(2), 115–134.

Sevilla, A., & Smith, S. (2020) Baby steps: the gender division of childcare during the COVID-19 pandemic, *Oxford Review of Economic Policy*, 36 (1), 169–186.

Stephan, U., Uhlaner, L. (2010). Performance-based vs socially supportive culture: A crossnational study of descriptive norms and entrepreneurship. *Journal of International Business Studies*, 41, 1347–1364.

Thébaud, S. (2010). "Gender and Entrepreneurship as a Career Choice: Do Self-Assessments of Ability Matter?" *Social Psychology Quarterly*, 73(2), 288-304.

Thébaud, S. (2015). Business as plan B: Institutional foundations of gender inequality in entrepreneurship across 24 industrialized countries. *Administrative science quarterly*, 60(4), 671-711.

Thébaud, S. (2016). Passing up the Job: The Role of Gendered Organizations and Families in the Entrepreneurial Career Process. *Entrepreneurship Theory and Practice*, 40(2), 269–287.

Tirole, J. (1988). *The Theory of Industrial Organization*. MIT Press.

Tirole, J. (2011). Illiquidity and All Its Friends. *Journal of Economic Literature*, 49(2), 287-325.

Torres, J., Maduko, F., Gaddis, I., Iacovone, L., & Beegle, K. (2023). The World Bank Research Observer, 38 (1), 36–72.

UNESCO. (2022). *Global monitoring of school closures.* Available at: <u>https://webarchive.unesco.org/web/20220629024039/https://en.unesco.org/covid19/ed</u> <u>ucationresponse/</u>

Urbano, D., Alvarez, C. (2014). Institutional dimensions and entrepreneurial activity: an international study. *Small Business Economics*, 42, 703–716.

Urbano, D., Aparicio, S. & Audretsch, D. (2019). Twenty-five years of research on institutions, entrepreneurship, and economic growth: what has been learned?. *Small Business Economics*, 53, 21–49.

Walker, E., Wang, C., & Redmond, J. (2008). Women and work-life balance: Is home-based business ownership the solution? *Equal Opportunities International*, 27(3), 258–275.

Welter, F., & Smallbone, D. (2011). Institutional Perspectives on Entrepreneurial Behavior in Challenging Environments. *Journal of Small Business Management*, 49, 107-125.

World Bank (2023a). Gender Data Portal.

World Bank (2023b). World Development Indicators.

10. Appendices

a. Figures



Figure 10.1: Number of days with fully closed schools, excluding school holidays, public holidays, and weekends

Source: Own elaboration, based on public data from OECD (2023).



Figure 10.2: Monthly number of registries of RES and the Official Journal (%)

Source: Own elaboration, based on public data from RES monthly reports from the Ministry of Economy, Development and Tourism.



Figure 10.3: Monthly number of registries of RES and the Official Journal, 2013-2023

Note: The dashed line indicates the month when the new AES was implemented (2020), that made all the registration steps available fully online.

Source: Own elaboration, based on public data from RES monthly reports from the Ministry of Economy, Development and Tourism.



Figure 10.4: Percentage of national population in each phase of the Step-by-Step Plan

Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to reopen. Source: Own elaboration, based on public data from the Ministry of Science, Technology, Knowledge and Innovation.



Figure 10.5: COVID-related conditions

Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to reopen. Source: Own elaboration, based on public data from the Ministry of Science, Technology, Knowledge and Innovation.



Figure 10.6: Percentage of students with open schools

Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to reopen. Source: Own elaboration, based on public data from the Ministry of Education.





Note: The first dashed line indicates the week when schools were ordered to close by the central government, while the second dashed line indicates the day which schools were officially allowed to reopen. Source: Own elaboration, based on public data from the Ministry of Education and the Ministry of Science, Technology, Knowledge and Innovation.



Figure 10.8: Step-by-Step Plan by municipality

Source: Own elaboration, based on public data from the Ministry of Education and the Ministry of Science, Technology, Knowledge and Innovation.



Figure 10.9: Percentage of students with open schools by municipality

Source: Own elaboration, based on public data from the Ministry of Education and the Ministry of Science, Technology, Knowledge and Innovation.



Figure 10.10: Men and women entrepreneurs by children's age

b. Tables

Table 6: Entrepreneurial response to school closures and reopen of biparental households with children ages 3–18. (School closure and reopen 2)

	Parents with children between 3-6 years	Parents with children between 7-14 years	Parents with children between 15-18 years
School closure	-0.000145**	-0.000101***	-0.0000958*
	(-3.17)	(-3.82)	(-2.55)
Woman	-0.000175***	-0.000166***	-0.000166***
	(-18.97)	(-30.79)	(-21.03)
School closure#Mother	-0.0000734***	-0.0000529***	-0.0000239*
	(-5.74)	(-7.29)	(-2.44)
School reopen	-0.000103*	-0.0000513	-0.0000475
	(-2.10)	(-1.82)	(-1.21)
School reopen#Mother	-0.0000447*	-0.0000464***	-0.0000267
	(-2.05)	(-3.93)	(-1.89)
Female pre-closure baseline mean	0.0001476	0.0001239	0.001169
Sociodemographic & AES controls	Yes	Yes	Yes
Municipal & Week FE	Yes	Yes	Yes
Phases & epidemiological controls	Yes	Yes	Yes
Observations	30,370,417	79,508,785	39,132,524

t statistics in parentheses

*p<0.05, ** p<0.01, *** p<0.001,

Table 7: Entrepreneurial response to school closures and reopen of biparental				
households by socioeconomic quintiles. (school closure and reopen 2).				

	Quintile I and			
	II	Quintile III	Quintile IV	Quintile V
School closure	-0.0000999**	-0.000112*	-0.0000641	-0.000158
	(3.05)	(2.09)	(0.92)	(1.57)
Woman	-0.000117***	-0.000150***	-0.000182***	-0.000319***
	(18.74)	(14.83)	(14.43)	(17.64)
School closure#Mother	-0.0000443***	-0.0000431**	-0.0000510***	-0.0000682*
	(6.18)	(3.04)	(3.52)	(2.57)
School reopen	-0.000049	-0.0000472	-0.0000504	-0.000176
	(1.37)	(0.83)	(0.66)	(1.69)
School reopen#Mother	-0.0000393***	-0.0000488*	-0.0000243	-0.0000353
	(3.9)	(2.45)	(1.08)	(0.96)
Female pre-closure baseline mean	0.00007809	0.000107	0.000146	0.000313
Sociodemographic & AES controls	Yes	Yes	Yes	Yes
Municipal & Week FE	Yes	Yes	Yes	Yes
Phases & epidemiological controls	Yes	Yes	Yes	Yes
Observations	58,466,747	19,120,474	15,312,239	13,236,202

t statistics in parentheses *p<0.05, ** p<0.01, *** p<0.001,